Author's response to reviews

Title: SnoN expression is differently regulated in microsatellite unstable compared with microsatellite stable colorectal cancers

Authors:

June A Chia (June.Chia@qimr.edu.au)
Lisa A Simms (Lisa.Simms@qimr.edu.au)
Sarah-Jane Cozzi (Sarah-Jane.Cozzi@qimr.edu.au)
Joanne Young (Joanne.Young@qimr.edu.au)
Jeremy R Jass (jeremy.jass@mcgill.ca)
Michael D Walsh (Michael.Walsh@qimr.edu.au)
Kevin J Spring (Kevin.Spring@qimr.edu.au)
Barbara A Leggett (Barbara_Leggett@health.qld.gov.au)
Vicki LJ Whitehall (Vicki.Whitehall@qimr.edu.au)

Version: 4 Date: 22 October 2006

Author's response to reviews:

Dear Sir / Madam,

Please find attached our revised manuscript entitled "SnoN expression is differently regulated in microsatellite unstable compared with microsatellite stable colorectal cancers" for resubmission to BMC Cancer. As requested, we have altered the manuscript to conform to the requirements of BMC Cancer. Specifically, we have:

1. Expanded the author list to include the full first name of each author
2. Used sentence case for section headings
3. Corrected our referencing of website addresses
4. Changed authors to authors'
5. Cropped our figure to minimise space surrounding the image

Thank you for accepting our manuscript for publication in BMC Cancer.

Kind regards,

Vicki Whitehall, PhD